Author's response to reviews

Title: Return-to-work of sick-listed workers without an employment contract - what works?

Authors:

Sylvia J. Vermeulen (s.vermeulen@vumc.nl)
Sietske J. Tamminga (s.j.tamminga@amc.uva.nl)
Ton A.J.M. Schellart (ton.schellart@vumc.nl)
Jan Fekke Ybema (jan_fekke.ybema@tno.nl)
Han J.R. Anema (h.anema@vumc.nl)

Version: 5 Date: 29 June 2009

Author's response to reviews:

Dear Editor,

On behalf of all authors, I hereby resubmit a new version of our manuscript: “Return-to-work of sick-listed workers without an employment contract – what works?”

All requested formatting changes have been addressed.

Major revisions:

Author list – The list of authors in the manuscript is now written exactly as the list in the submission system.

Footnotes – The footnote form has been removed from the title page. The footnotes are incorporated in the text. The affiliations are written beneath the author list.

Authors’ contributions – The sentence ‘all authors read and approved the final manuscript’ has been added.

Minor revisions:

E-mail addresses – This subheading is removed from the title page.

Keywords – These have been removed.

Figure and table placement text – These have been removed.

Tables – Vertical lines are removed.

Table title and legend – This point has been addressed. Table titles are placed above the tables.

Underlining – This has been changed to bold, as suggested.
Figure cropping – This point has been addressed. The cropped figures are uploaded in the submission system.

With kind regards,

Sylvia J. Vermeulen, MD, MSc
Research Center for Insurance Medicine AMC-UWV-VUmc
VU University Medical Center – EMGO institute
P.O. Box 7057
1007 MB Amsterdam
The Netherlands